The impact of opiate substitution treatment on mortality risk in drug addicts: a natural experiment study

Colin D Steer,¹* John Macleod,¹ Kate Tilling,¹ Aaron G Lim,¹ John Marsden,² Tim Millar,³ John Strang,² Maggie Telfer,⁴ Heather Whitaker,⁵ Peter Vickerman¹ and Matthew Hickman¹

 ¹Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK
²National Addiction Centre, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
³Centre for Mental Health and Safety, School of Health Sciences, University of Manchester, Manchester, UK

⁴Bristol Drug Project, Bristol, UK

⁵Department of Mathematics and Statistics, The Open University, Milton Keynes, UK

*Corresponding author Colin.Steer@bristol.ac.uk

Declared competing interests of authors: John Marsden acknowledges research grants from the Department of Health and Social Care, the National Institute for Health Research (NIHR) and the NIHR Biomedical Research Centre for Mental Health at South London and Maudsley NHS Mental Health Foundation Trust, and part-time employment as Senior Academic Advisor for the Alcohol, Drugs and Tobacco Division, Health and Wellbeing Directorate, Public Health England. He declares investigator-led educational grant funding from Indivior PLC (administered by Action on Addiction) for a study of adjunctive, personalised psychosocial intervention for non-response to opioid agonist treatment (ARC Trial), and support from NIHR (Health Technology Assessment) for a trial of extended-release naltrexone. He has received honoraria from Merck Serono (Darmstadt, Germany; 2015; clinical oncology medicine); Martindale Pharma (Brentwood, UK; 2017; treatment for opioid use disorder); and Indivior PLC (via PCM Scientific) as co-chairperson and chairperson (2015–18) for the conference on Improving Outcomes in Treatment of Opioid Dependence. Tim Millar has received research funding from the UK National Treatment Agency for Substance Misuse, Public Health England and the Home Office. He has been a member of the organising committee for conferences supported by unrestricted educational grants from Reckitt Benckiser Group plc (Slough, UK), Lundbeck Ltd (Milton Keynes, UK), Martindale Pharma and Britannia Pharmaceuticals Ltd (Reading, UK), for which he received no personal remuneration. He is a member of the UK Advisory Council on the Misuse of Drugs. John Strang is a clinician and researcher and has worked extensively with agencies in the addiction treatment fields and addiction-related charities and with government departments and has contributed to clinical guidelines on treatment types and provision. John Strang's employer (King's College London) has received, connected to his work, project grant support and/or honoraria and/or consultancy payments from the Department of Health and Social Care, the National Treatment Agency, Public Health England, the Home Office, the National Institute for Health and Care Excellence and the European Monitoring Centre for Drugs and Drug Addiction (EMCDDA), as well as research grants from (2016–18) NIHR, the Medical Research Council and the Pilgrim Trust. He has also worked with the World Health Organization, the United Nations Office on Drugs and Crime, EMCDDA, the US Food and Drug Administration and the US National Institute on

Drug Abuse, as well as other international government agencies. John Strang's employer (King's College London) has also received, connected to his work, research grant support and/or payment of honoraria, consultancy payments and expenses from pharmaceutical companies [including, for 2016–18, Martindale, Indivior PLC, Mundipharma (Cambridge, UK) and Braeburn/Camurus (Lund, Sweden)] and trial medication supply from iGen and Braeburn. John Strang's employer (King's College London) has registered intellectual property on an innovative buccal naloxone with which John Strang has been named in a patent registration for concentrated naloxone nasal spray by Euro-Celtique SA on behalf of Mundipharma Research Limited. For updated information see www.kcl.ac.uk/ioppn/depts/addictions/people/hod.aspx (accessed 2 October 2018). Matthew Hickman has received unrelated unrestricted honoraria from Gilead Sciences, Inc. (Cambridge, UK), AbbVie (Maidenhead, UK), Janssen Pharmaceuticals (High Wycombe, UK) and Merck Serono. Matthew Hickman is a member of the Public Health Research Research Funding Board.

Published January 2019 DOI: 10.3310/hsdr07030

Plain English summary

The impact of opiate substitution treatment on mortality risk Health Services and Delivery Research 2019; Vol. 7: No. 3 DOI: 10.3310/hsdr07030

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Users of heroin or other opioids, such as morphine, have a risk of death 10 times higher than that of the general population. Overdose is the most common cause of death. In England and Wales during 2015/16, over 1200 people died from opioid poisonings, the largest number on record. The most effective treatment for people dependent on opioids is the prescription of substitute drugs, usually methadone or buprenorphine, called opiate substitution treatment. In the UK, this treatment is delivered commonly in primary care, often with support from drug agency workers.

We analysed data from people on opiate substitution treatment in primary care. We assessed whether or not death rates (all-cause and overdose) change with different periods of treatment or between treatments, with buprenorphine compared with methadone, and with the co-prescription of other drugs such as benzodiazepines.

Mortality risk was lowest after 4 weeks of treatment, at 0.3% for overdose deaths, but it was eight times higher in the first 4 weeks after treatment ceased. There was evidence that mortality risk was lower for patients on buprenorphine than for those on methadone, especially in the first 4 weeks of treatment, when the mortality risk for the former was approximately 90% lower.

The co-prescription of benzodiazepines more than doubled overdose death rates. The co-prescription of other drugs (zopiclone and similar sedatives, and gabapentinoids) also increased the overdose risk by 60%.

Higher doses in the first 4 weeks of treatment may be associated with higher death rates. If patients dropped out of treatment rather than having their dose gradually lowered, this might have led to higher death rates in the first 4 weeks after treatment ceased.

The number of buprenorphine prescriptions per year has increased over time, but treatment duration is shorter for patients on buprenorphine than for those on methadone.

New interventions are required that retain patients on treatment in the community. Clinical guidance on the dangers of co-prescribing drugs with opiate substitutes may need strengthening.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Steer *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme or programmes which preceded the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The Health Services and Delivery Research (HS&DR) programme, part of the National Institute for Health Research (NIHR), was established to fund a broad range of research. It combines the strengths and contributions of two previous NIHR research programmes: the Health Services Research (HSR) programme and the Service Delivery and Organisation (SDO) programme, which were merged in January 2012.

The HS&DR programme aims to produce rigorous and relevant evidence on the quality, access and organisation of health services including costs and outcomes, as well as research on implementation. The programme will enhance the strategic focus on research that matters to the NHS and is keen to support ambitious evaluative research to improve health services.

For more information about the HS&DR programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hsdr

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 12/136/105. The contractual start date was in September 2013. The final report began editorial review in October 2017 and was accepted for publication in July 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the NHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Steer *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk